Author’s response to reviews

Title: Use of sugammadex in patients with neuromuscular disorders: a systematic review of case reports

Authors:
Usha Gurunathan (usha.gurunathan@health.qld.gov.au)
Shakeel Meenan Kunju (shakeel.kunju@health.qld.gov.au)
Lisa Stanton (lisa.stanton@health.qld.gov.au)

Version: 4 Date: 01 Nov 2019

Author’s response to reviews:

Dear Editor

We have made recommended changes to the manuscript as follows:

1) Author's contributions: The contribution of author L.S has been more clearly mentioned in the methods section and in the declarations sections. Please let us know if further information is required.

2) Two statements added to the declaration section: ethics & consent and consent for publication, as advised.

3) Tables 2-5 have already been cited in the subheadings of the neuromuscular diseases in the discussion section. However, as requested, we have cited along the body of the discussion as well.

4) With table 6, as requested, we have introduced the endnotes section before the references section. We have removed 1-2 sentences from table 6 suit the section, and hence edited the references. We have also converted list into a paragraph. Please let us know if this format is sufficient.

5) Abstract headings have been changed as requested.

6) PRISMA checklist has been removed from inventory

7) Clean copy has been uploaded.

Please inform us if any more change is required.

Kind regards